Author's response to reviews

Title: Are decision trees a feasible knowledge representation to guide extraction of critical information from randomized controlled trial reports?

Authors:

Grace Y Chung (gigigraceyc@gmail.com)
Enrico Coiera (e.coiera@unsw.edu.au)

Version: 4 Date: 12 October 2008

Author's response to reviews:

Dear Editor,

Please find attached the latest version with formatting revisions. As requested the following revisions were made.

1. All superscripted references have been converted as requested.
2. Underline has been removed as requested.
3. All the figures are already cropped as close as possible.

Please let me know if further revisions are needed.

Regards,
Grace Chung.